Author’s response to reviews

Title: Comparison of efficacy between incretin-based therapies for type 2 diabetes mellitus

Authors:

Kaustubh Nisal (knisal@gmail.com)
Ram Kela (drramkela@doctors.org.uk)
Kamlesh Khunti (kk22@le.ac.uk)
Melanie Davies (melanie.davies@uhl-tr.nhs.uk)

Version: 7 Date: 11 October 2012

Author’s response to reviews:

Dear Editorial team

Many thanks for your time

I have revised the manuscript as per your suggestions.

1. The manuscript is now submitted as a word format document
2. All track changes been removed
3. Coloured text is removed and back colour is used.
4. Tables have been incorporated into the main document
5. Additional files removed.

Many thanks

Yours sincerely

Kaustubh Nisal